Author's response to reviews

Title: Prevalence of childhood disability and the characteristics and circumstances of disabled children in the UK: Secondary analysis of the Family Resources Survey

Authors:

Clare M Blackburn (c.m.blackburn@warwick.ac.uk)
Nick J Spencer (n.j.spencer@warwick.ac.uk)
Janet M Read (j.m.read@warwick.ac.uk)

Version: 4 Date: 16 March 2010

Author's response to reviews:

Dear Editor,

Thank you for this opportunity to revise and resubmit our paper to your journal.

As requested in your email of 15 March, we have revised the manuscript as follows:

i) Tables have been reformatted so they now fit onto the width of an A4 portrait page
ii) Abbreviations have been included in a paragraph rather than a list
iii) All vertical visible lines have been removed from tables.
iv) Checked for typographical areas and a few minor corrections made.